Author's response to reviews


Authors:

Beverley M Essue (beverley.essue@sydney.edu.au)
Maree L Hackett (mhackett@georgeinstitute.org.au)
Andreas Mueller (amueller@hollows.org)
Nguyen Thi Hanh Duc (nguyenhanhduc@hollows.org.vn)
Huynh Tan Phuc (huynhtanphuc@hollows.org.vn)
Stephen Jan (sjan@georgeinstitute.org.au)

Version: 3 Date: 23 August 2011

Author's response to reviews:

23 August 2011

Dear BMC Series Editorial Production Team,

Thank you for the opportunity to review the final proof of this protocol and to address the additional issues raised. Below I have outlined the issues raised by your team and how I have addressed each issue:

Abstract: this abstract does not have the correct structure. Please reformat it to use these headings: Background, Methods/Design, Discussion - complete

Structure: this manuscript does not have the correct structure. Please reformat it to use these headings:

Abstract; Background; Methods/Design; Discussion; Abbreviations (if any); Competing interests; Authors' contributions; Authors' information (optional); Acknowledgements (optional); References - complete. Please note, within the Methods/Design section, we have used sub-headings to improve flow and readability of the protocol. Please clarify if sub-headings are acceptable. If not, I will remove them.

References: Please make sure references are cited in numerical order - corrected

References: please provide a functioning link to website http://www.wpro.who.int/countries/2009/vtn/ - complete

Acknowledgements: Please place the "Study management and funding" in an Acknowledgements section directly proceeding the references - complete

Tables- please remove the visible vertical lines from your tables - complete
Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures - complete

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance - complete.

We trust that these responses address all of the comments raised. Please let us know if you require any additional information to process this publication. We look forward to viewing this manuscript online once it has been published. Please contact me on either +61 2 9993 4562 or beverley.essue@sydney.edu.au if you require additional information.

Sincerely,

Beverley Essue

Co-authors: Maree L Hackett, Andreas Mueller, Nguyen Thi Hanh Duc, Huynh Tan Phuc and Stephen Jan